

Automatic Expansion of Domain-Specific Affective Models for Web Intelligence Applications

Albert Weichselbraun^{1,4} · Jakob Steixner² · Adrian M.P. Braşoveanu² · Arno Scharl^{3,4} · Max Göbel⁴ · Lyndon J. B. Nixon^{2,3}

Received: 27 June 2020 / Accepted: 12 January 2021 / Published online: 30 January 2021 © The Author(s) 2021

Abstract

Sentic computing relies on well-defined affective models of different complexity—polarity to distinguish positive and negative sentiment, for example, or more nuanced models to capture expressions of human emotions. When used to measure communication success, even the most granular affective model combined with sophisticated machine learning approaches may not fully capture an organisation's strategic positioning goals. Such goals often deviate from the assumptions of standardised affective models. While certain emotions such as *Joy* and *Trust* typically represent desirable brand associations, specific communication goals formulated by marketing professionals often go beyond such standard dimensions. For instance, the brand manager of a television show may consider *fear* or *sadness* to be desired emotions for its audience. This article introduces expansion techniques for affective models, combining common and commonsense knowledge available in knowledge graphs with language models and affective reasoning, improving coverage and consistency as well as supporting domainspecific interpretations of emotions. An extensive evaluation compares the performance of different expansion techniques: (i) a quantitative evaluation based on the revisited *Hourglass of Emotions* model to assess performance on complex models that cover multiple affective categories, using manually compiled gold standard data, and (ii) a qualitative evaluation of a domain-specific affective model for television programme brands. The results of these evaluations demonstrate that the introduced techniques support a variety of embeddings and pre-trained models. The paper concludes with a discussion on applying this approach to other scenarios where affective model resources are scarce.

Keywords Affective models · Hourglass of emotions · Language models · Embeddings · Knowledge graphs

	Albert Weichselbraun albert.weichselbraun@fhgr.ch; weichselbraun@weblyzard.com
	Jakob Steixner steixner@modultech.eu
	Adrian M.P. Brașoveanu adrian.brasoveanu@modul.ac.at; brasoveanu@modultech.eu
	Arno Scharl arno.scharl@modul.ac.at; scharl@weblyzard.com
	Max Göbel goebel@weblyzard.com
	Lyndon J. B. Nixon nixon@modultech.eu; lyndon.nixon@modul.ac.at
1	University of Applied Sciences of the Grisons, Chur, Switzerland
2	MODUL Technology, Vienna, Austria
3	MODUL University Vienna, Vienna, Austria
4	webLyzard technology, Vienna, Austria

Introduction

Organisations use Web intelligence applications to obtain real-time insights into the public perception of their brands. Driven by news media coverage, influential social media postings and real-world events, the mood of consumers and their perception of a brand can change rapidly. Consumers who discuss brands through digital channels not only respond to communication, but also play a pivotal role in shaping brand reputation, for example when repeating or commenting on a story. This reflects their personal connection to the brand and the collective nature of brand authorship. Sentiment and affective categories are important indicators derived from these user actions. They help organisations to better understand the public debate and track evolving perceptions of their brands. To measure communication success, however, general emotional categories often do not suffice. Domain-specific affective models incorporate specific emotional categories that are not found in general models. Their interpretation might also deviate from typical perceptions. (In the case of a television show, for example, *fear* or *sadness* may represent desirable associations.) Another advantage of domainspecific affective models is the possibility to include not only emotional categories but also other desired or undesired semantic associations specific to the situational context.

To provide actionable knowledge for communication professionals, Web intelligence applications should therefore support both: (i) standardised affective models to benchmark multiple brands and compare the results with third-party studies and (ii) domain-specific affective models that consider the specific communication goals of an organisation. Often formulated in an ad hoc manner, e.g. during a communications workshop, the major challenge of such models relates to their inconsistent and often incomplete nature. They tend to have low coverage since the time and effort invested into their definition and disambiguation cannot compete with standardised affective models based on many years of scientific research.

Addressing this problem, this article introduces a datadriven method to expand domain-specific affective models in situations when lexical resources required as training data are scarce. The method automatically extends such models through knowledge graph concepts in conjunction with language models and affective reasoning. The goal is to improve the coverage and consistency of affective models such as the *webLyzard Stakeholder Dialogue and Opinion Model* (WYSDOM),¹ which provides a communication success metric that combines sentiment and emotional categories with desired and undesired semantic associations. Computed based on co-occurrence patterns, these associations provide real-time insights into the success of marketing and public outreach activities.

WYSDOM goes beyond sentiment and standardised emotional categories by asking communication experts to specify the intended positioning of their organisations. This positioning is expressed in the form of desired and undesired keywords. In the case of the U.S. National Oceanic and Atmospheric Administration (NOAA), for example, an association with "climate change" in the public debate indicates successful communication although the term typically carries a negative sentiment [1].

Tracking the WYSDOM metric over time allows to assess to what extent a chosen communication strategy impacts the public debate, how consistently a message is being conveyed and whether this message helps to reinforce the intended brand positioning in a sustainable manner. The visual representation of the metric comes is a stacked bar chart that combines content-based metrics (positive vs. negative sentiment and desired vs. undesired associations) with other indicators of success such as page views and the number of visits (see the lower half of Fig. 1). While the metric has initially been created for tracking the success of brand communication, it is applicable to a wide range of use cases that benefit from a hybrid display of content-based metrics in conjunction with other *Key Performance Indicators* such as sales figures or stock market prices.

We have investigated several affective models for possible inclusion of their emotional categories into the WYSDOM metric, as shown in Fig. 1. These models include *Sentiment* (Positive, Neutral, Negative), *Brand Personality* (Sincerity, Competence, Ruggedness, Sophistication, Excitement and the added dimension Sustainability) according to Aaker [2], Plutchik's *Wheel of Emotions* [3], as well as Cambria et al.'s *Hourglass of Emotions* [4]. The latter provides a comprehensive multidimensional framework for interpreting emotions used in diverse fields such as ontology construction [5] and affective visualisation [6].

This paper showcases a method to augment domainspecific affective models with concepts extracted from open knowledge graphs such as ConceptNet [7] and WordNet [8], lexical resources, pre-trained embeddings like Global Vectors (GloVe) [9], domain documents from multiple sources and language models such as the Bidirectional Encoder Representations from Transformers (BERT) and its distilled version known as DistilBERT [10]. It provides a fast and reliable method to build domain-specific affective classifiers even if resources required for the tasks are scarce.

The rest of the article is organised as follows: Section 2 and Section 3 provide an overview of affective models and related work. Section 4 introduces the affective model expansion method, as well as the affective knowledge extraction method used for the evaluations. Section 5 presents the gold standard (Section 5.1) and discusses two use cases: (i) a quantitative evaluation that expands an affective model based on the revised Hourglass of Emotions (Section 5.2) and (ii) a qualitative evaluation that demonstrates the method's suitability for improving domainspecific affective models (Section 5.3). Section 6 discusses the evaluation results. The concluding Section 7 summarises the contribution and highlights the strengths and weaknesses of the presented method.

Affective Models

This section first provides background information on sentiment and emotion classification models. It then outlines the relation of these models to work on domain-specific affective models.

¹ www.weblyzard.com/wysdom-success-metric



Fig. 1 Overview of affective models including Sentiment, Brand Personality, Wheel of Emotions, Hourglass of Emotions and the webLyzard Stakeholder Dialogue and Opinion Model (WYSDOM)

Sentiment and Emotion Classification Models

Sentiment analysis aims at determining whether a statement is positive (e.g. 'Awesome battery life.'), negative (e.g. 'Do not waste your money on this phone!'), neutral (e.g. 'I bought this for my spouse.') or ambivalent (e.g. mixes multiple polarities, 'Great display but horrible battery life.'). Emotion analysis, in contrast, provides a much more fine-grained classification by recognising the emotion(s) expressed in a text and mapping them to emotional categories.

Research in this area has significantly gained traction in recent years, developing classification models that draw on psychology [11, 12], neuroscience [12], social science [13], computer science [14] and engineering [15, 16]. The complexity of emotion processing has led to many definitions and interpretations that differ in the specific aspects considered: physiological processes, evolutionary adaptation to environmental stimuli, affective evaluation or the subjectivity of emotional experience [17]. This variety is reflected in an extensive review of affective models and algorithms by Wang et al. [18], which covers nine popular models and 65 emotions discussed in these models.

Many authors have formulated classification systems of emotions such as Ekman's basic emotions [19]. Some are well-known in business and marketing, such as the Wheel of Emotions [3], the Circumplex Model of Affect [20] or the Hourglass of Emotions [4] and its revised version [12]. The structure of these frameworks often distinguishes basic and derived emotions, e.g. *envy* derived from the combination of *shame* and *anger*.

The Hourglass of Emotions is a comprehensive and multidimensional framework for interpreting emotions, inspired by neuroscience and motivated in psychology. The initial version [4] distinguished four affective categories: pleasantness (defined as joy-sadness, based on the affective concepts in this category), attention (anticipation-surprise), sensitivity (anger-fear) and aptitude (trust-disgust). By using the different activation scales (e.g. pleasantness can have different activation levels characterised as ecstasy, joy, serenity, pensiveness, sadness and grief) and composition, the model can express a wide range of emotions. The Hourglass of Emotions has also been used extensively in information visualisation due to its colour associations. Recently, a revised version [12] was published that further improved the original model by removing neutral emotions, increasing consistency (e.g. comfort and discomfort are now classified as opposites), and adding polar emotions as well as self-conscious emotions. The model also refined the colour scheme to be in line with recent studies on colour-emotion associations and considerably improved the polarity scores obtained for compound emotions. The revised model changes the definition of the four primary affective categories and the associated affective concepts into introspection (joy-sadness), temper



Fig.2 Types of affective models including possible mappings, as indicated by the dotted lines

(*calmness-anger*), attitude (*pleasantness-disgust*) and sensitivity (*eagerness-fear*), while also providing additional explanations on how to create compound emotions using the new emotion classification scheme. Some researchers might consider the elimination of *surprise* problematic, but in terms of classification it is a welcome improvement since most annotators found it difficult to decide whether it should be considered a positive or negative emotion.

Domain-Specific Affective Models

Sentiment polarity and emotion categories often do not coincide with desired business outcomes such as the brand perception that is aspired for or an organisation's public relations goals. Scharl el al. [1] have therefore introduced the WYSDOM success metric that allows companies to define domain-specific affective models. These models aim at capturing affective content relevant to their specific business communication goals.

We consider "affective models" as an umbrella term that covers sentiment polarity, standard affective models using common emotion categorisations as well as domain-specific affective models. Each model covers different affective dimensions (e.g. sentiment polarity or the emotions defined by the specific affective model) and might even provide mappings to and from other models. SenticNet 6, for instance, translates text into primitives and subsequently superprimitives from which it inherits a specific set of emotions that in turn can be mapped to a particular polarity [21].

Figure 2 outlines the relation between three types of affective models: (i) Sentiment polarity is the best understood model from a research perspective. It only includes one affective dimension to distinguish positive, neutral and negative documents. (ii) Emotion *classification* models that consider multiple affective dimensions are more challenging, which is reflected in the large number of models from different authors. (iii) Domain-specific affective models follow a more customised approach, relating affective content to the specific communication goals of an organisation. They typically focus on a small number of affective dimensions, given their customised nature and the manual effort involved in creating the model's specifications and continuously updating them in line with evolving communication goals. The dotted lines in the figure indicate possible mappings between these models, for



Fig.3 Affective analysis of the media coverage on the COVID-19 pandemic based on Plutchik's *Wheel of Emotions*, using a tag cloud (left), a radar chart (middle) and keyword graph (right) with colour coding to distinguish selected emotions including Anticipation, Vigilance, Fear, Anger and Sadness



Fig. 4 Screenshot of the webLyzard Web intelligence dashboard with results of a query on *World Health Organisation* between April and June 2020—using colour coding to visualise the emotions *Anticipation, Fear, Trust, Acceptance* and *Interest*

example from emotions to sentiment polarity as outlined by Cambria et al. [12].

Figure 3 shows a previous application of affective analysis conducted as part of the *Corona Mood Barometer*.² The system uses a combination of story detection and emotion analysis techniques to discover what drives the public coronavirus 2019 (COVID-19) debate and how government responses to the coronavirus pandemic are perceived across the various countries. The visualisations explore associations with five selected emotions (Anticipation, Vigilance, Fear, Anger and Sadness) to better understand the drivers of the public debate in May 2020. The tag cloud sorts associations alphabetically, colour coding them by emotion. The radar chart projects the top keywords along multiple axes, revealing the relative strength of association with each emotional category. The keyword graph then applies a hierarchical layout, with grey centre nodes to represent keywords linked to multiple emotional categories.

Figure 4 shows how the visualisations can be accessed via the webLyzard Web intelligence dashboard, based on a

search query for the *World Health Organization* (WHO) that resulted in more than 14,000 documents published between April and June 2020. The dashboard is an advanced information exploration and retrieval interface that helps to track the various emotions along multiple context dimensions (sources, regions, languages, etc.) and enables on-the-fly filtering and query refinement options to access a comprehensive content repository of news and social media content.

Related Work

The discussion of related work starts with deep learning techniques for classifying affective categories such as emotions, then provides an overview of methods used for extracting content-based communication success metrics from news and social media, and a discussion of how Natural Language Processing (NLP) can support this process. Additional background information on this area of research is available in recent surveys such as those published in Cambria et al. [22], Xing et al. [23], Chaturvedi et al. [24] and Mehta et al. [25].

² www.weblyzard.com/corona-mood-barometer

Deep Learning for Affective Classification

During the last decade, the paradigm of affective models shifted due to the rise of language models based on deep learning methods that capture the meaning of all the words from a text corpus as a set of vectors in a lower-dimensional space (e.g. embeddings like word2vec [26], GloVe [9], and fastText [27]). With the addition of attention mechanisms and Transformer architectures [28], current language models such as BERT [29] were shown to be better at picking up linguistic phenomena [30] and performing tasks as diverse as capturing analogies, semantic role labelling, textual entailment, sentiment analysis and named entity recognition (NER). NER is particularly important for processes like coreference and anaphora resolution [31]. Coreference resolution refers to the process of finding all expressions that refer to the same object or entity and linking them to a single identifier, whereas anaphora resolution is the process through which the antecedent of an expression is determined. These tasks are important for affective reasoning as they support subjectivity detection [24], which helps us understand whether a text refers to the subject or is the subject's opinion on a product or review. Subjectivity *detection* [24] is a complex problem that has moved through various stages from manually crafted features and bootstrapping to syntactic features and domain adaptation via knowledge graphs and neural networks and finally to cross-modal fusion of text, video and audio via Transformer models like BERT. While hand-crafted features produced many false positives, classic machine learning (ML) with syntactic features missed even shallow representations of meaning, which were later added by knowledge graphs. Combining subjectivity detection with advanced filtering mechanisms such as threat or sarcasm detection [32], cause-pairs extraction [33] or concept-level sentiment analysis [34] enables a fine-grained approach towards affective classification and provides support for operations like removal of bias or propaganda, provision of better context awareness, and better accuracy of subjectivity values. The next trend in subjectivity detection research seems to be focused on distinguishing cultural aspects, biases, nuances and dialects.

Affective classification is usually framed as a text classification task, as outlined by Kowsari et al. [35] and Wolf et al. [10] who surveyed deep learning architectures used for sentiment analysis. A multi-layer perceptron (MLP) stacked ensemble is used for predicting emotional intensity for different content types such as Twitter postings, microblogs and news [36]. It showed significant improvements over similar systems for both generic emotion analysis and financial sentiment analysis. The EvoMSA [37] open-source multilingual toolkit for creating sentiment classifiers composes the outputs of multiple models (fastText, Emoji Space, lexicon-based model, etc.) into a vector space that is then wired into the EvoDAG genetic algorithm to predict the final class. The performance improvements brought by EvoMSA are impressive, but the resulting models obtained from the EvoDAG algorithm were not designed to be explainable.

The key element of successful NLP language models is a process called knowledge transfer, which refers to the transfer of learnt patterns (e.g. weights) from one problem to another [38]. In some cases, if there is a need to reduce the size of the models and a small reduction in performance is acceptable, it is also possible to use a process called knowledge distillation [10], a lightweight knowledge transfer process for compressed models that are smaller and faster. Due to the knowledge transfer process, Transformer models such as BERT pick up various linguistic phenomena like direct objects, noun modifiers and coreferents [30], which benefits their performance in tasks like sentiment analysis and NER [29].

Another key research problem is the adaptation of affective models to new domains. This is necessary since many domains introduce special terminology or jargon, which can lead to misinterpretations of the affective categories they convey. In contrast to the work introduced in this paper, domain adaptation techniques do not create domain-specific affective models but rather fine-tune existing models (e.g. sentiment polarity). Xing et al. [39] present a cognitive-inspired adaptation method that emulates metacognition processes for detecting contradictions and obtaining the correct sentiment polarity of words when a human is confronted with a new language domain. Murtadha et al. [40] use weak supervised methods to cluster words according to their sentiment polarity for aspect-based sentiment analysis in the target domains. They also apply an attentionbased long short-term memory (LSTM) network to the same task. Both models work well due to the weight reduction for non-sentiment parts from a sentence. Zhao et al. [41] discuss multi-source domain adaptation for a cross-domain sentiment classification task. Their paper shows that joint learning for cross-domain tasks leads to good results and a greater generalisation capability, while at the same time enabling deep domain fusion. Domain adaptation techniques can also vary depending on the architecture. Since Transformers like BERT are generally task-agnostic, the best method to adapt them to another domain is optimising training by applying strategies such as adversarial training, pre-training and post-training [42].

Extraction of Content-Based Communication Success Metrics from Web and Social Media

The extraction of communication success metrics from digital content streams is a dynamic research area that relies heavily on NLP and information visualisation. Traditionally, sentiment is among the most frequently used metrics for evaluating the impact of a campaign. The required computation can therefore be formulated as sentiment polarity extraction (e.g. the identification or positive or negative emotions) or stance classification (e.g. the classification of opinions towards a certain target) [42]. The task of determining the polarity of the considered resources typically leverages lexical resources that map numerical values to terms with an affective meaning (e.g. the value -1 for terms on the negative affective scale such as *fear*, or +1 for positive terms such as *trust*). Lexical resources for this purpose show different levels of granularity, e.g. mere polarity vs. subjectivity vs. fine-grained aspects.

Related analytic tasks focus on event discovery and the representation of relations as a knowledge graph, as presented by Nguyen et al. [43] and Camacho et al. [44]. The integration of knowledge graphs with ML approaches aids the recognition of emotions in languages such as Arabic and Spanish for a wide variety of NLP tasks such as knowledge transfer and machine translation [44]. Sentic computing [45] helps to model categories such as life satisfaction and safety, because it provides linguistic cues on emotions such as *sadness, joy, anger* and *fear*. Regardless of the computational methods used, the goal of the sentiment analysis is to produce values stemming from differences between evaluative ratings of positive and negative emotions.

Well-known ontologies for sentiment analysis include OntoSenticNet [5] and the multilingual visual sentiment ontology [46]. OntoSenticNet [5] acts as a commonsense ontology for the sentiment domain. The visual sentiment ontology [46] is used in the multimedia domain. Since such ontologies often cover a limited number of domains, a semiautomatic ontology builder was proposed for solving the issue of domain adaptation for aspect-based sentiment analysis [47]. In addition to ontologies, knowledge graphs are used to support complementary tasks like entity detection and commonsense reasoning. DBpedia [48] and Wikidata [49] are public knowledge graphs typically used in entity linking tasks, whereas ConceptNet [7] and SenticNet [21] are used in sentiment and emotion detection. The last version of SenticNet used subsymbolic artificial intelligence (e.g. clustering, recommendation algorithms) to detect patterns in natural language and represent them with symbolic logic in a knowledge graph. The key to understanding the construction of the latest version of SenticNet is the idea of language composition, namely the fact that multi-word expression can be deconstructed into primitives or superprimitives (e.g. functions that will be able to represent an entire range of primitives). To compute the polarity, it suffices to look up the value of the superprimitives associated with the respective primitive.

Many papers are dedicated to the construction of domainspecific affective knowledge graphs for aspect-based analysis. One method to create such graphs is showcased in Cavallari et al. [50] and is based on graph embeddings, namely the embedding of entire communities instead of individual nodes. This leads to significant improvement in applications like community detection or node classification. Another method for constructing knowledge graphs is presented by Ghosal et al. [51]: the filtering of ConceptNet to create a domain-aggregated graph that is then fed to a graph convolutional network (GCN) autoencoder to build a domain-adversarial training dataset, which includes both domain-specific and domain-agnostic concepts. Du et al. [42] leverage learnt entity and relation embeddings to fully exploit the constraints of a commonsense knowledge graph. Bijari et al. [52] combine sentence-level graph-based learning representations with latent and continuous features extraction to improve sentiment polarity detection. Custom affective graphs are built in order to provide both word sense disambiguation and affective reasoning for specific domains like law and medicine (e.g. K-BERT [53]) and finance (e.g. FOREX market prediction [54]). Finally, a recent survey expands upon the problems of building, representing and applying knowledge graphs for affective reasoning [55].

Method

Many content-based success metrics consider affective categories like sentiment polarity, emotions and domainspecific metrics such as (un)desired keyword associations, e.g. for the above-mentioned WYSDOM model. They require multifaceted sentic computing engines that integrate syntactics (e.g. part-of-speech tagging, chunking, lemmatisation), semantics (e.g. topic extraction, named entities) and *pragmatics* (e.g. sarcasm detection, aspect extraction) layers [22]. The authors have developed components across all these layers, including an aspect-based sentiment analysis engine [56], a named entity linking (NEL) engine [57] and a NLP and visualisation pipeline that includes topic, concept and story detection [58]. These components are used as a basis for the extraction of social indicators. The computed sentiment values typically include aspect, polarity and subjectivity. Aspect is used to analyse the various features of products or ideas, polarity offers the details about sentiment orientation (positive, neutral or negative), and subjectivity describes a person's opinion towards a product, topic or idea.

Developing the method to expand domain-specific affective models has been guided by the following goals and constraints:

- The method should be applicable in both research and corporate settings, covering different types of affective models (comprehensive standardised affective lexicons as well as tailored domain-specific models such as WYSDOM).
- 2. The expansion process should not require large and comprehensive corpora, since creating such corpora is not feasible in most industrial settings.
- The expansion process must perform well across domains and should draw on publicly available resources such as common and commonsense knowledge, pre-trained word embeddings and language mod-



Fig. 5 Semantic enrichment, contextualisation and affective reasoning based on language models and word embeddings for affective knowledge expansion

els to improve (by disambiguating terms based on their context) and expand (by extending the affective lexicon) affective models.

4. The expanded affective models should be usable with simple lexicon-based sentiment analysis techniques as well as with more sophisticated approaches that consider syntactical (i.e. negation, modifiers, quotes, etc.) and contextual (i.e. disambiguation of the term based on its actual use in a sentence) information.

Section 4.1 introduces the affective model expansion method. Section 4.2 then presents an affective knowledge extraction technique that builds on the expanded models and considers the sentence's grammar and context in the extraction process.

Affective Model Expansion

The affective model expansion technique uses explicit knowledge available in lexical databases and knowledge graphs such as WordNet [8], ConceptNet [7] and Wikidata [49] as well as implicit knowledge about a term's semantics encoded in word embeddings and language models. Figure 5 outlines the iterative affective model expansion process in greater detail. The method enriches terms and phrases from the seed model with structured knowledge obtained from publicly available sources such as WordNet, ConceptNet and Wikidata by mining synonyms, antonyms and phrases that are related to the seed terms.

The next step aims at contextualising the enriched model by mining WordNet and domain corpora for sentences that contain concepts from the affective model. Contextualisation does not only facilitate the use of language models such as BERT and DistilBERT, but also allows improving the precision and consistency of the affective model by splitting ambiguous terms into multiple concepts (senses). The example sentences that demonstrate a term's use in a context are transformed into the embedding space and disambiguated based on Algorithm 1, which is part of the affective reasoning component. Table 1 provides examples that illustrate the outcome of this process based on the ambiguous terms *like*, *probe* and *project* with two selected senses for each term. The left side of the table shows the affective categories assigned to the terms' average senses, while the right-hand columns present the affective categories assigned to the disambiguated senses.

Algorithm 1: Disambiguates a term (t) with
the affective categories (s) by computing a dic-
tionary s_dict of all the term's senses and the
corresponding affective categories.
Data: t, s_t
Result: s dict[sense]
1 senses $\leftarrow get$ senses(t);
2 sense vector dict $\leftarrow \{\}$;
3 s dict \leftarrow {};
/* compute the centroid of the term's senses
and the average distance between senses. */
4 foreach sense in senses do
5 usage examples \leftarrow get example(sense);
6 usage vectors $\leftarrow \ln(\text{usage examples})$;
τ sense_vector_dict[sense] \leftarrow avg(usage_vectors);
8 end
$9 \text{ sense_centroid} \leftarrow \operatorname{avg}(\operatorname{sense_vector_dict});$
10 avg_sense_dist \leftarrow avg_dist(sense_vector_dict);
/* validate affective categories of senses. */
11 foreach sense in senses do
12 if (cos(sense_vector_dict[sense], sense_centroid
$\leq avg_sense_dist \cdot 1.3$) then
13 $s_dict[sense] \leftarrow s_t$;
14 end
15 else
16 $s_dict[sense] \leftarrow$
get_affective_categories(sense);
17 end
18 end
19 returns dict.

For terms that are available in WordNet, the algorithm loops over all senses, retrieves examples of each sense and uses the language model to transform them into the corresponding embedding space for that particular sense. Algorithm 1 then computes a centroid that represents the term's average usage and computes the senses' average distance from this centroid. Finally, we assign the senses' overall values for all semantic categories to senses that are close to the term's average usage and compute a refined set of semantic categories for terms that are different from the seed term's average usage. For terms not covered in WordNet, Algorithm 1 is modified to use example sentences mined from the domain corpus rather than on explicit WordNet senses.

Once all multi-sense terms have been successfully disambiguated, the affective reasoning component performs a proximity search as outlined in Algorithm 2 to further expand the affective model. The expanded model may then run through another iteration to enrich the extracted concepts, contextualise them and transform all terms into embedding space.

Algorithm 2: Proximity lookup and affective reasoning heuristic used for computing the semantic categories (sc) of a new term (t) based on the language model lm.

Data: t, lm, affective_categoriesResult: s1 $s \leftarrow \{\}$;2 $pt \leftarrow get_proximate_terms(lm, t)$;3 $ant \leftarrow identify_antonyms(lm, pt)$;4 $sim \leftarrow pt - ant$;5 foreach ac in affective_categories do6avg_similar \leftarrow avg_affective_value(sim, ac);7 $avg_antonyms \leftarrow avg_affective_value(ant, ac);$ 8 $s[ac] \leftarrow avg(avg_similar, -avg_antonyms);$ 9 end10 return s;

Affective Knowledge Extraction

Figure 6 provides an overview of the affective knowledge extraction method underlying the experiments discussed in Section 5.2. The component uses the expanded and contextualised affective models by transforming the input sentence into embedding space and then applying semantic reasoning to all sentence tokens. The use of Transformer language models such as BERT also considers the token's concept, i.e. disambiguating the concept prior to determining its values alongside the affective categories. In addition, dependency parsing and grammar rules

provide information on the token's grammatical context, which is useful for considering negation and modifiers that determine a token's impact on the sentence's affective categories.

The affective knowledge extraction computes a feature vector based on (i) the token (t_i) , (ii) the corresponding sentence (s_i) and (iii) the sentence's dependency tree (dp_i) .

Equation 1 computes the sentence score along the affective category (ac) by first obtaining the *embedding_i* based on token (t_i) and the context information available in sentence (s_j) from the chosen language model (lm). The algorithm then uses a proximity search and the approach outlined in Algorithm 2 to determine the value of the affective category in the context of the sentence. Finally, we compute the score for the affective category, considering negation and modifiers based on the dependency tree (dp_j) with the factors $n(dp_j, i)$ and $m(dp_j, i)$ respectively, as shown in Equation 2.

$$embedding_i = lm(t_i, s_j) \tag{1}$$

$$score(s_j, ac) = \sum_{i=1}^{n} m(dp_j, i) \cdot n(dp_j, i) \cdot score(embedding_i, ac)$$
(2)

All algorithms were run using an in-house tokeniser to seamlessly integrate the presented approach with other components in our Web intelligence platform (e.g. text clean-up, keyword and topic extraction [59], dependency parsing [60] and NEL [57]). For classic embeddings (e.g. word2vec, GloVe) we have used the gensim³ library, whereas for the BERT and DistilBERT models we have used wrappers on top of the Spacy⁴ [61] and Transformers⁵ [10] libraries.

Evaluation

The evaluation process aimed at providing both quantitative insights into the methods' performance and qualitative results that support the quantitative assessment.

The quantitative evaluation in Section 5.2 is based on the revised *Hourglass of Emotions* model. For benchmarking the affective model expansion and affective knowledge extraction components, we created a gold standard

³ www.radimrehurek.com/gensim

⁴ www.spacy.io

⁵ https://github.com/huggingface/transformers



Fig. 6 Using language models and the expanded affective models for affective knowledge extraction

comprising 346 sentences, annotated according to the affective categories defined in the model (see Section 5.1).

The qualitative evaluation focuses on domain-specific affective models used as the basis for computing the WYSDOM communication success metric. The selected model stems from the European Horizon 2020 Research Project ReTV, *Re-Inventing TV for the Digital Age*,⁶ which develops knowledge extraction and visualisation services for broadcasters and media archives. The experiments illustrate how the custom model has gained in extent and consistency due to the application of the method introduced in Section 4.

Revisited Hourglass of Emotions Gold Standard

The annotation process started with a corpus of over 12,000 sentences extracted from Wikinews. To ensure that the annotators had formed a good understanding of the revised Hourglass of Emotions model, we created a set of annotation rules and selected 500 sentences for annotation, separating some of them for testing purposes and using the rest for the gold standard. The annotation rules were collected in the *annotation guideline*, which followed rules similar to sentic evaluation challenges (e.g. SemEval [62], SMM4H [63] or WASSA

[64]). For each class (*introspection, temper, pleasantness* and *eagerness*), a set of rules was provided to guide the annotators.

The annotators were first asked to read the guidelines and to comment on them. After agreeing on the rules and their interpretation, the human experts annotated the documents using the (i) annotation guideline and (ii) tables explaining the revisited Hourglass of Emotions categories from [12]. Providing these tables improved the quality and consistency of the classifications considerably, as it helped avoiding incorrect classifications of emotions as None or Unknown. In addition, the annotation guideline contained about 50 triggers for each polar opposite of the affective categories. Selected from the dictionaries, the triggers provided cues for the annotators (e.g. a set of triggers like awe, force, malady, defeat, terror, danger, flood, violence that point the annotators towards emotions like anxiety, fear and terror representing the negative polarity for sensitivity). A set of examples of each polar opposite of an affective category was also selected from previously published corpora including Saravia et al. [65] and Poria et al. [66] as well as the extracted Wikinews sentences. The list of triggers and examples of emotional categories eventually helped to distinguish subtle nuances.

Table 1Selected exampleswith two senses per term for the
contextualisation of seed terms
based and the corresponding
values assigned to the affective
categories (T)emper, (I)
ntrospection (A)ttitude and (S)
ensitivity for these terms

term	senses	contextualised example	Т	Ι	А	S
like	a similar kind	we don't want the likes of you around here	0.00	0.45	0.00	0.00
	wish, care, like	Would you like to come along to the movies?	0.00	0.49	0.00	0.57
probe	investigation	there was a congressional probe into the scandal	0.62	-0.56	-0.43	0.62
	poke into	probe an anthill	0.00	0.00	0.00	0.00
project	communicate vividly	He projected his feelings	0.00	0.00	-0.76	0.42
	throw, send	project a missile	0.00	-0.44	0.00	-0.56

⁶ www.retv-project.eu

The human annotators originated from different cultures, and the gender ratio was balanced. Selected documents were annotated twice. Annotators evaluated 120 sentences each, providing information about the sentence's affective categories (i.e. emotions that occur in the sentence), the dominant emotion and the overall polarity of the sentence. In the case of ambiguous examples, annotators highlighted the sentence and provided their reasoning behind the choice. *Unknown* was assigned to cases where it was not possible to classify the dominant emotion, and *None* to sentences not containing any emotional expressions.

Finally, an expert with relevant experience in sentic computing validated the gold standard annotations. The final gold standard was created only after a consensus between the expert and annotators had been found for difficult cases.⁷ The dominant emotion was determined by the expert based on the individual annotator assessments, whereas the affective categories and polarity were averaged. Around 70% of the selected sentences were included in the final corpus (346 sentences out of 500 initially selected), with high inter-rater agreement (Fleiss kappa=0.868). Anger (the negative polarity of the temper category) and the category None turned out to be the major sources of disagreement. Although the remaining 30% of the sentences with no associated affective category were not used in the evaluation, they were kept in the corpus. Disagreements related to Anger were often related to cultural differences in annotator perceptions. Table 2 summarises annotator agreement across classes (e.g. the four affective categories of the revised Hourglass of Emotions model split by polarity plus the None class). The fact that no sentences were marked as Unknown supports the assumption that the revised Hourglass of Emotion model is well-suited for classifying emotions [12].

Table 2 Agreement (Fleiss kappa) within the gold standard for positive and negative values of the affective categories (*A*)titude, (*I*)ntrospection, (*S*)ensitivity and (*T*)emper used in the revisited Hourglass of Emotions model. The *None* category indicates sentences with no emotion

category	agreement	
A+	pleasantness	0.90
A-	disgust	0.86
I+	joy	0.90
I-	sadness	0.85
S+	eagerness	0.88
S-	fear	0.97
T+	calmness	0.85
Т-	anger	0.78
None	-	0.78

Quantitative Analysis Based on the Revisited Hourglass of Emotions Model

The presented evaluations use the F1 score, which is defined as the harmonic mean of precision and recall. The experiments performed in this section compute the evaluation scores based on two approaches:

- 1. The performance metric based on the *dominant affective category* determines whether the dominant emotion corresponds to the one presented in the gold standard that yields a True Positive (TP). Otherwise, the sentence is considered a False Positive (FP). The obtained score equates to the recall for the dominant emotion.
- 2. The second metric determines whether *all affective categories* present in a sentence have been correctly detected. If the affective category in the gold standard matches the one computed by the system, we obtain a

Table 3Recall of the dominantemotion based on the affectivevalues encoded in SenticNet 5and extensions: plain indicatesresults based on the applicationof the SenticNet 5 lexicon asa static lookup table, +ARsignifies affective reasoning,+L lemmatisation, and +GRgrammar rules

category		plain	+AR	+AR+L	+GR	+AR+GR	+AR+L+GR
T+	calmness	0.50	0.64	0.61	0.50	0.68	0.64
T-	anger	0.22	0.39	0.47	0.24	0.47	0.56
I+	joy	0.63	0.66	0.80	0.60	0.69	0.80
I-	sadness	0.37	0.57	0.57	0.43	0.61	0.61
A+	pleasantness	0.82	0.80	0.84	0.76	0.78	0.80
A-	disgust	0.49	0.61	0.50	0.47	0.58	0.54
S+	eagerness	0.46	0.62	0.54	0.56	0.64	0.56
S-	fear	0.50	0.67	0.57	0.47	0.63	0.57
overall		0.50	0.62	0.60	0.49	0.63	0.63

⁷ The gold standard and annotation rules are available at the following address: https://github.com/modultechnology/affective-models-corpora.

Table 4 Precision/recall/F1 from direct application of the GloVe model on SenticNet 5. Note that despite the already large size of the lexicons, applying affective reasoning (+AR) significantly improves the results

category		plain	+AR	+AR+L	+GR	+AR+GR	+AR+L+GR
T+	calmness	0.31/0.55/0.40	0.34/0.58/0.43	0.31/0.5/0.38	0.29/0.50/0.30	0.38/0.63/0.48	0.34/0.53/0.42
Т-	anger	0.58/0.26/0.36	0.64/0.40/0.50	0.63/0.4/0.49	0.54/0.26/0.30	0.70/0.46/0.55	0.67/0.47/0.55
I+	јоу	0.55/0.74/0.63	0.62/0.77/0.69	0.62/0.85/0.71	0.56/0.74/0.64	0.64/0.80/0.71	0.63/0.85/0.72
I-	sadness	0.69/0.45/0.55	0.75/0.63/0.69	0.79/0.58/0.67	0.71/0.49/0.58	0.78/0.64/0.71	0.80/0.60/0.69
A+	pleasantness	0.53/0.75/0.63	0.61/0.78/0.68	0.57/0.83/0.67	0.52/0.71/0.60	0.61/0.79/0.69	0.58/0.79/0.67
A-	disgust	0.71/0.41/0.52	0.73/0.54/0.62	0.75/0.46/0.57	0.66/0.41/0.51	0.73/0.54/0.62	0.73/0.51/0.61
S+	eagerness	0.48/0.46/0.47	0.60/0.57/0.58	0.56/0.52/0.54	0.47/0.43/0.45	0.60/0.59/0.59	0.58/0.54/0.56
S-	fear	0.59/0.53/0.56	0.67/0.69/0.68	0.62/0.62/0.62	0.58/0.53/0.55	0.67/0.67/0.67	0.64/0.64/0.64
overall		0.58/0.50/0.54	0.64/0.62/0.63	0.63/0.58/0.61	0.57/0.50/0.53	0.66/0.64/0.65	0.64/0.61/0.63

True Positive (TP). For affective categories that have not been detected, the metric yields False Negatives (FN). False Positives (FP) are returned for computed categories that are not present in the gold standard. The second metric corresponds to the F1 score, listed together with precision and recall.

The first evaluations aim at quantifying the improvements from affective model expansion, based on the revisited Hourglass of Emotions as seed model. In contrast to SenticNet 5, which covers 100,000 commonsense concepts and had already been published at the time our experiments have been conducted, SenticNet 6 has only become available after June 2020 [21]. Therefore, the extension of an affective model based on the revisited Hourglass of Emotions model [12] has been the perfect candidate for testing our approach:

- It describes a sophisticated affective model that can be used in conjunction with the gold standard developed in Section 5.1 to design reproducible experiments for quantifying the impact of the expansion process on model performance, and
- Stakeholders can benefit from the expanded model since it allows extracting affective knowledge on

emotions based on the categories defined in the revisited Hourglass of Emotions model.

Evaluation Based on SenticNet 5

The first set of evaluations based on SenticNet 5 do not yet consider the improvements of the SenticNet 6 model [21] nor extensions to this model based on the method introduced in Section 4.1. The SenticNet 5 lexicon covers many n-grams that do not necessarily have precomputed vectors in the GloVe model and are thus not annotated. This still leaves several tens of thousands of single-token terms with non-zero values for each affective dimension.

Tables 3 and 4 summarise the outcome of these experiments. The performance of a similarity-based approach that considers affective reasoning (+AR) is compared against a static lookup based on the same initial term–value lexicon. The similarity calculation is based on the *glove-wiki-gigaword-300* pre-trained model. The results indicate that the suggested expansion considerably improves the performance across all SenticNet categories. The first column (plain) describes the evaluation outcome based on the unmodified SenticNet 5 lexicon, the second column (+AR) provides the results after the model expansion, and the final column (+AR+L) shows the expanded model based on lemmas rather than the unmodified terms. All

Table 5Recall of the dominantemotion using the GloVelanguage model on SenticNet5, with and without applyingdependency parsing andgrammar rules (GR)

category		plain	+AR	+AR+L	+GR	+AR+GR	+AR+L+GR
T+	calmness	0.00	0.39	0.39	0.00	0.61	0.61
T-	anger	0.06	0.71	0.71	0.06	0.78	0.75
I+	joy	0.17	0.66	0.63	0.17	0.66	0.63
I-	sadness	0.04	0.72	0.67	0.07	0.80	0.76
A+	pleasantness	0.07	0.73	0.75	0.07	0.75	0.71
A-	disgust	0.06	0.72	0.72	0.07	0.72	0.71
S+	eagerness	0.05	0.59	0.59	0.05	0.62	0.62
S-	fear	0.07	0.73	0.67	0.07	0.80	0.67
overall		0.06	0.67	0.66	0.07	0.72	0.69

Table 6 Precision/recall/F1 for a model based on the revisited Hour-	rea
glass of Emotions with simple word embeddings (GloVe): plain uses the	cat
small generated dictionaries for static lookup, +AR indicates affective	

reasoning for matching novel terms, +L lemmatisation, and +GR application of grammar rules/negation/dependency parsing

category		plain	+AR	+AR+L	+GR	+AR+GR	+AR+L+GR
T+	calmness	0.00/0.00/0.00	0.50/0.37/0.42	0.50/0.37/0.42	0.33/0.03/0.05	0.72/0.61/0.66	0.68/0.61/0.64
Т-	anger	0.90/0.13/0.22	0.70/0.79/0.75	0.70/0.76/0.73	1.00/0.13/0.22	0.80/0.86/0.83	0.79/0.81/0.80
S+	eagerness	0.02/1.00/0.04	0.65/0.57/0.60	0.62/0.57/0.59	1.00/0.07/0.12	0.63/0.59/0.61	0.63/0.59/0.61
S-	fear	0.33/0.04/0.07	0.67/0.75/0.71	0.66/0.69/0.67	0.33/0.04/0.07	0.67/0.71/0.69	0.67/0.69/0.68
A+	pleasantness	0.50/0.06/0.11	0.72/0.66/0.69	0.72/0.68/0.70	0.56/0.06/0.12	0.76/0.70/0.73	0.74/0.68/0.71
A-	disgust	0.55/0.07/0.12	0.73/0.77/0.75	0.72/0.74/0.73	0.54/0.08/0.14	0.75/0.80/0.77	0.73/0.77/0.75
I+	joy	0.64/0.11/0.18	0.73/0.71/0.72	0.70/0.69/0.70	0.78/0.11/0.19	0.76/0.74/0.75	0.73/0.72/0.73
I-	sadness	0.60/0.04/0.08	0.50/0.77/0.76	0.74/0.74/0.74	0.71/0.07/0.13	0.77/0.79/0.78	0.77/0.77/0.77
overall		0.58/0.07/0.12	0.68/0.67/0.68	0.68/0.66/0.67	0.65/0.07/0.13	0.74/0.73/0.73	0.72/0.71/0.71

three experiments use the SenticNet 5 lexicon as the basis, and they only differ in the additional logic applied during the annotation phase. While the inclusion of grammar parsing (+GR) notably improves the result, this is not the case for lemmatisation.

Since annotators were instructed to annotate the gold standard sentences with the affective categories of the revisited Hourglass of Emotions model in mind, somewhat lower results are to be expected. The evaluation used the following mapping between the affective categories in the revisited model (left) and the ones in SenticNet 5 (right) with their respective poles:

- sensitivity (eagerness/fear): sensitivity (anger/fear)
- attitude (*pleasantness/disgust*): aptitude (*trust/disgust*)
- introspection (*joy/sadness*): pleasantness (*joy/sadness*)
- temper (calmness/anger): attention (anticipation/surprise)

The mapping from temper to attention has proven to be the most problematic one yielding the lowest metrics in Tables 3 and 4.

Evaluation Based on the Hourglass of Emotions

The second experiment drew upon the affective categories used in the revisited Hourglass of Emotions (*temper*, *introspection*, *attitude* and *sensitivity*) and example concepts taken from [12]. For affective categories already present in the original SenticNet model, we selected the top 20 terms, ignoring illnesses, from SenticNet 5. For new categories we manually added additional terms to provide a more balanced seed set. Tables 5 and 6 illustrate the impact of the affective model expansion process on the model's performance, which considerably improves recall for all affective categories.

The first interesting observation is model performance before the expansion, which is considerably lower when compared to SenticNet 5. This was expected given the limited size of the seed model. It only covers 445 affective concepts in total, as compared to 300-500 affective concepts per affective category in the expanded lexicons. Consequently, many sentences are assigned a neutral value and most nonneutral sentences are affected by a single trigger. SenticNet 5 has a considerably higher coverage although its affective categories differ from the revised model. This is illustrated in model performance after the expansion process, which

Table 7Recall of the dominantemotion for a model basedon the revisited Hourglass ofEmotions using the BERT/DistilBERT language model,with and without applyingdependency parsing andgrammar rules (GR)

category		BERT	DistilBERT	BERT+GR	DistilBERT+GR
Г+	calmness	0.62	0.46	0.75	0.68
Г-	anger	0.55	0.65	0.45	0.65
i+	joy	0.37	0.46	0.40	0.43
[-	sadness	0.76	0.80	0.74	0.83
4+	pleasantness	0.62	0.64	0.65	0.67
4-	disgust	0.68	0.69	0.69	0.68
8+	eagerness	0.38	0.36	0.46	0.36
S-	fear	0.90	0.87	0.80	0.70
overall		0.61	0.63	0.62	0.64

Table 8Precision/recall/F1values for all emotions withina sentence using the BERT/DistilBERT language modelwith and without applyingdependency parsing andgrammar rules (GR) on a modelthat has been based on therevisited Hourglass of Emotions

category		BERT	DistilBERT	BERT+GR	DistilBERT+GR
T+	calmness	0.45/0.61/0.52	0.50/0.39/0.44	0.52/0.76/0.62	0.71/0.63/0.67
T-	anger	0.75/0.57/0.65	0.72/0.71/0.71	0.87/0.57/0.69	0.84/0.74/0.79
I+	joy	0.66/0.35/0.46	0.73/0.42/0.53	0.65/0.43/0.52	0.71/0.38/0.50
I-	sadness	0.60/0.79/0.69	0.65/0.84/0.73	0.63/0.77/0.69	0.66/0.81/0.73
A+	pleasantness	0.62/0.55/0.58	0.66/0.61/0.64	0.65/0.60/0.62	0.69/0.58/0.63
A-	disgust	0.65/0.64/0.65	0.71/0.68/0.69	0.69/0.64/0.67	0.73/0.70/0.72
S+	eagerness	0.71/0.43/0.54	0.69/0.39/0.50	0.68/0.50/0.58	0.63/0.39/0.47
S-	fear	0.64/0.85/0.73	0.64/0.85/0.73	0.67/0.82/0.74	0.61/0.76/0.68
overall		0.64/0.58/0.61	0.67/0.60/0.63	0.70/0.61/0.65	0.68/0.62/0.65

yields an almost five-fold improvement, even more than in the experiment described in Section 5.2.1. The inclusion of grammar rules (GR) further improves the results.

Tables 7 and 8 outline performance gains achieved by applying Transformer-based language models such as BERT and DistilBERT. Several other models were also tested (e.g. RoBERTa, XLNet) but since the classic BERT model (*bert-base-uncased*) and the distilled model (*distillbert-base-uncased*) yielded the best initial results, we have only considered these in the final evaluations.

Qualitative Evaluation Based on a Domain-Specific Affective Model

As part of the requirements elicitation process in the ReTV project, domain experts from the participating media organisations were asked to provide short lists of desired and undesired associations for their organisations. Their input was condensed into the ten undesired and ten desired concepts listed in Table 9 (column *seed model*), which were then fed into the affective model expansion process. This process yielded the additional concepts listed in the column *expanded model*.

As in the experiments from the previous section, the expanded model considerably improved recall. A qualitative analysis performed on Wikinews articles showed that the suggested model extensions allowed the identification of additional affective content that would have been classified as neutral with the seed model. Table 10 lists example sentences, their corresponding desirability score and affective concepts that have been identified based on the extended model.

Discussion

The experiments were designed to establish a baseline for the proposed method's performance on real-world affective models. The method can be used in conjunction with a wide variety of embeddings, from classic approaches such as word2vec and GloVe to more recent embeddings extracted from Transformer language models like BERT. Pre-trained and unmodified versions of GloVe, BERT and DistilBERT yielded significant improvements of the evaluated models. This is encouraging given that publicly available models were used without further optimisations. The suggested approach works well for models of varying complexity and in settings where lexical resources are scarce, as has been the case for the revisited Hourglass of Emotions model.

Results could be further improved by using custom embeddings or fine-tuning the selected language model. In such an optimised setting, BERT and DistilBERT are likely to outperform GloVe, which yielded the best results for the pre-trained models (Table 6). BERT was not necessarily built to solve all classes of NLP problems. While it has shown good results for tasks like entity recognition or basic sentiment analysis, serious issues have surfaced during evaluations of fine-grained inference problems. As shown by Ettinger [67], BERT had shortcomings in regard to the impact of negation within larger contexts. This was the main reason to develop models for improving

 Table 9
 Selected concepts from the seed model (left) and the expanded affective model (right) for the broadcasting domain

-		0	
seed model desired	undesired	expanded model desired	undesired
balanced captivating entertaining informative innovative investigative professional reliable transparent trustworthy	boring censored disrespectful fake irrelevant offensive partisan self-referential unbalanced unprofessional	articulate businesslike captivating competent dependable dispassionate enlightening entertaining insightful unbiased	callous convoluted degrading demeaning groundless intolerant misguided pointless slanderous undignified

title	desirabality	affective concepts
"Just to make this perfectly clear, I was laughing at the joke and not at any group of people."	1.0	just: 1.0; make: 1.0; perfectly clear: 1.0;
		laughing: -0.05; group: -0.05
So while coverage for Democrats overall was a bit more positive than negative, that was almost all due to extremely favorable coverage for Obama.	1.0	coverage: 0.05; overall: 0.2; bit: 0.05; positive:1.0;
		negative: -1.0; extremely favorable: 1.0
The statement drew criticism to the network for being false.	-1.0	criticism: -0.55; false: -0.7
Jeet Heer, the national affairs correspondent at The Nation said "the big loser of the night was the network that hosted the event."	-0.715	affairs: -0.05; loser: -1; Nation said: -0.05

Table 10 Positive (desired) and negative (undesired) results obtained with the expanded affective model

BERT's negation handling, like NegBERT [68], as negation detection is important not only for clinical text analysis, but also for bias, sarcasm or the general task of affective inference. Since we are not using NegBERT-based models, our approach complements this work and can be applied on top of existing BERT models constrained by negation issues.

Table 8 sheds light on another interesting aspect: results for positive examples trail those for negative examples, regardless of method and model. This outcome suggests a negative bias within the Wikinews gold standard, which is confirmed by the literature noting that most political articles have a negative connotation [69]. Dependency parsing and rule-based syntactic processing can yield improved results, even for sophisticated language models that extract contextualised embeddings for documents in the pipeline.

Table 9 illustrates that the proposed method can help increase the explainability of affective models, since it is possible to inspect and modify the seed models and the expanded affective models, whereas Table 10 shows some positive and negative results obtained with the expanded affective model.

Future research could focus on better understanding the linguistic information encoded in the resulting models, e.g. by using structural probing [70], structured perceptron parsers [71] or visualisations—as demonstrated through BERT embeddings and attention layers visualisations like those from [30] and [72].

Outlook and Conclusion

Advanced Web intelligence applications should be able to provide domain-specific communication success metrics tailored to an organisation's evolving communication goals. Such metrics depend on affective models that measure and classify how a brand, product or service is perceived across digital channels.

This article introduces a novel method for improving the coverage and consistency of such affective models that combines the advantages of word embeddings with the robustness of lexical approaches and knowledge graphs. It provides a flexible, fast and inexpensive method to create and expand affective models. The expressiveness of the method can be controlled by the complexity of the chosen embeddings. A quantitative evaluation confirmed that the expanded models clearly outperform the original models, even in resource-scarce scenarios. To conduct this evaluation, we have created a gold standard of Wikinews sentences that was annotated with the affective categories defined in the revised 2020 version of the *Hourglass of Emotions* model [12].

Integrating the results into WYSDOM combines the strength of established and extensively evaluated affective models with the ability to create domain-specific models on the fly, based on a limited set of desired and undesired associations that can be elicited in a single workshop with the communication experts responsible for a brand or product.

Future work to further advance this approach will focus on: (i) improving the affective reasoning components by incorporating techniques to cluster related word senses rather than using fixed, empirically chosen threshold values, (ii) incorporating additional common and commonsense knowledge sources into the expansion process, and (iii) adapting the language models to the affective model's domain.

Acknowledgements The authors would like to thank Katinka Böhm, Adriana Bassani and Lenka Kilian for their help in the specification of emotional categories and their participation in the quantitative evaluation. Funding Information Open Access funding provided by the University of Applied Sciences of the Grisons (www.fhgr.ch). This research has been partially funded through the following projects: the MedMon project (www.fhgr.ch/medmon) funded by Innosuisse (No. 25587.2 PFES-ES); the ReTV project (www.retv-project.eu) funded by the European Union's Horizon 2020 Research and Innovation Programme (No. 780656) and the EPOCH project (www.epoch-project.eu) funded by the Austrian Federal Ministry for Climate Action, Environment, Energy, Mobility and Technology (BMK) via the ICT of the Future Program (GA No. 867551).

Compliance with ethical standards

Conflicts of interest The authors declare that they have no conflict of interest and that informed consent was obtained from all evaluation participants.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons. org/licenses/by/4.0/.

References

- Scharl A, Herring DD, Rafelsberger W, Hubmann-Haidvogel A, Kamolov R, Fischl D, Föls M, Weichselbraun A. Semantic systems and visual tools to support environmental communication. IEEE Syst J. 2017;11(2):762–71. https://doi.org/10.1109/JSYST. 2015.2466439.
- 2 Aaker JL. Dimensions of brand personality. J Mar Res. 1997;34(3):347-56. https://doi.org/10.2307/3151897.
- Plutchik R. A General Psychoevolutionary Theory of Emotion. In Theories of emotion. Elsevier, 1980. p. 3–33. https://doi. org/10.1016/B978-0-12-558701-3.50007-7.
- Cambria E, Livingstone A, and Hussain A. The Hourglass of Emotions. In Cognitive behavioural systems, Springer 2012. p. 144–157. https://doi.org/10.1007/978-3-642-34584-5_11.
- 5 Dragoni M, Poria S, Cambria E. OntoSenticNet: A Commonsense Ontology for Sentiment Analysis. IEEE Intell Syst. 2018;33(3):77-85. https://doi.org/10.1109/MIS.2018.033001419.
- 6 Kucher K, Paradis C, Kerren A. The State of the Art in Sentiment Visualization. Comput Graphics Forum. 2018;37(1):71–96. https://doi.org/10.1111/cgf.13217.
- Speer R, Chin J, and Havasi C. ConceptNet 5.5: An Open Multilingual Graph of General Knowledge. In Proceedings of the Thirty-First AAAI Conference on Artificial Intelligence, February 4-9, 2017, San Francisco, California, USA 2017. p. 4444–4451. http://aaai.org/ocs/index.php/AAAI/AAAI17/paper/view/14972.
- 8 Miller GA, Fellbaum C. WordNet Then and Now. Lang Resour Eval. 2007;41(2):209–14. https://doi.org/10.1007/s10579-007-9044-6.
- Pennington J, Socher R, and Manning CD. Glove: Global Vectors for Word Representation. In Proceedings of the 2014 Conference

on Empirical Methods in Natural Language Processing, EMNLP 2014, October 25-29, 2014, Doha, Qatar, A meeting of SIGDAT, a Special Interest Group of the ACL, 2014. p. 1532–1543. https://doi.org/10.3115/v1/d14-1162.

- Wolf T, Debut L, Sanh V, Chaumond J, Delangue C, Moi A, Cistac P, Rault T, Louf R, Funtowicz M, and Brew J. HuggingFace's Transformers: State-of-the-art Natural Language Processing. CoRR, abs/1910.03771, 2019. http://arxiv.org/ abs/1910.03771.
- 11 Soares AP, Comesaa M, Pinheiro AP, Simes A, Frade CS. The Adaptation of the Affective Norms for English Words (ANEW) for European Portuguese. Behav Res Methods. 2012;44(1):256– 69. https://doi.org/10.3758/s13428-011-0131-7.
- 12 Susanto Y, Livingstone A, Ng BC, Cambria E. The Hourglass Model Revisited. IEEE Intell Syst. 2020;35(5):96–102. https:// doi.org/10.1109/MIS.2020.2992799.
- 13 Reeck C, Ames DR, Ochsner KN. The Social Regulation of Emotion: An Integrative, Cross-Disciplinary Model. Trends Cogn Sci. 2016;20(1):47–63. https://doi.org/10.1080/02699939208411068.
- Shivhare SN, Garg S, and Mishra A. EmotionFinder: Detecting Emotion from Blogs and Textual Documents. In International Conference on Computing, Communication & Automation, 2015. p. 52–57. https://doi.org/10.1109/CCAA.2015.7148343.
- Li H, Pang N, Guo S, and Wang H. Research on Textual Emotion Recognition Incorporating Personality Factor. In 2007 IEEE International Conference on Robotics and Biomimetics (ROBIO) 2007. p. 2222–2227. https://doi.org/10.1109/ROBIO.2007.4522515.
- Huangfu L, Mao W, Zeng D, and Wang L. OCC Model-Based Emotion Extraction from Online Reviews. In 2013 IEEE International Conference on Intelligence and Security Informatics 2013. p. 116–121. https://doi.org/10.1109/ISI.2013.6578799.
- 17 Lajante M, Ladhari R. The Promise and Perils of The Peripheral Psychophysiology of Emotion in Retailing and Consumer Services. J Retail Consum Serv. 2019;50:305–13. https://doi. org/10.1016/j.jretconser.2018.07.005.
- Wang Z, Ho SB, and Cambria E. A Review of Emotion Sensing: Categorization Models and Algorithms. Multimedia Tools and Applications, Jan. 2020. https://doi.org/10.1007/s11042-019-08328-z.
- Ekman P. An Argument for Basic Emotions. Cognition and Emotion. 1992;6(3–4):169–200. https://doi.org/10.1080/02699939208411068.
- 20 Posner J, Russell JA, Peterson BS. The circumplex Model of Affect: An Integrative Approach to Affective Neuroscience, Cognitive Development, and Psychopathology. Dev Psychopathol. 2005;17(3):715–34. https://doi.org/10.1017/S0954579405050340.
- Cambria E, Li Y, Xing FZ, Poria S, and Kwok K. SenticNet 6: Ensemble Application of Symbolic and Subsymbolic AI for Sentiment Analysis. In CIKM '20: The 29th ACM International Conference on Information and Knowledge Management, Virtual Event, Ireland 2020. p. 105–114. https://doi.org/10.1145/3340531.3412003.
- 22 Cambria E, Poria S, Gelbukh AF, Thelwall M. Sentiment Analysis Is a Big Suitcase. IEEE Intell Syst. 2017;32(6):74–80. https://doi. org/10.1109/MIS.2017.4531228.
- 23 Xing FZ, Cambria E, Welsch RE. Natural Language Based Financial Forecasting: A Survey. Artif Intell Rev. 2018;50(1):49–73. https://doi.org/10.1007/s10462-017-9588-9.
- Chaturvedi I, Cambria E, Welsch RE, Herrera F. Distinguishing Between Facts and Opinions for Sentiment Analysis: Survey and Challenges. Information Fusion. 2018;44:65–77. https://doi. org/10.1016/j.inffus.2017.12.006.
- 25 Mehta Y, Majumder N, Gelbukh AF, Cambria E. Recent Trends in Deep Learning Based Personality Detection. Artif Intell Rev. 2020;53(4):2313–39. https://doi.org/10.1007/s10462-019-09770-z.
- Levy O, and Goldberg Y. Linguistic regularities in sparse and explicit word representations. In Morante and Yih p. 171–180. https://doi.org/10.3115/v1/w14-1618.

- Joulin A, Grave E, Bojanowski P, and Mikolov T. Bag of Tricks for Efficient Text Classification. In Proceedings of the 15th Conference of the European Chapter of the Association for Computational Linguistics, EACL 2017, Valencia, Spain, April 3-7, 2017, Volume 2: Short Papers 2017. p. 427–431. https://doi. org/10.18653/v1/e17-2068.
- Vaswani A, Shazeer N, Parmar N, Uszkoreit J, Jones L, Gomez AN, Kaiser L, and Polosukhin I. Attention is All You Need. In Advances in Neural Information Processing Systems 30: Annual Conference on Neural Information Processing Systems 2017, 4-9 December 2017, Long Beach, CA, USA 2017. p. 5998–6008. http://papers.nips.cc/paper/7181-attention-is-all-you-need.
- 29. Devlin J, Chang M, Lee K, and Toutanova K. BERT: Pre-Training of Deep Bidirectional Transformers for Language Understanding. In Proceedings of the 2019 Conference of the North American Chapter of the Association for Computational Linguistics: Human Language Technologies, NAACL-HLT 2019, Minneapolis, MN, USA, June 2-7, 2019, Volume 1 (Long and Short Papers) 2019. p. 4171–4186.
- Chi EA, Hewitt J, and Manning CF. Finding Universal Grammatical Relations in Multilingual BERT. In Proceedings of the 58th Annual Meeting of the Association for Computational Linguistics, ACL 2020, Online, July 5-10, 2020, p. 5564–5577. https://www. aclweb.org/anthology/2020.acl-main.493/.
- Zhong X, Cambria E, Hussain A. Extracting Time Expressions and Named Entities with Constituent-Based Tagging Schemes. Cogn Comput. 2020;12(4):844–62. https://doi.org/10.1007/s12559-020-09714-8.
- 32. Chauhan DS, Ekbal DSRA, and Bhattacharyya P. Sentiment and Emotion Help Sarcasm? A Multi-task Learning Framework for Multi-Modal Sarcasm, Sentiment and Emotion Analysis. In Proceedings of the 58th Annual Meeting of the Association for Computational Linguistics, ACL 2020, Online, July 5-10, 2020. p. 4351–4360. https://www.aclweb.org/anthology/2020. acl-main.401/.
- Xia R, and Ding Z. Emotion-Cause Pair Extraction: A New Task to Emotion Analysis in Texts. In Proceedings of the 57th Conference of the Association for Computational Linguistics, ACL 2019, Florence, Italy, July 28- August 2, 2019, Volume 1: Long Papers, 2019. p. 1003–1012. https://doi.org/10.18653/v1/p19-1096.
- Satapathy R, Singh A, and Cambria E. PhonSenticNet: A Cognitive Approach to Microtext Normalization for Concept-Level Sentiment Analysis. In Computational Data and Social Networks 8th International Conference, CSoNet 2019, Ho Chi Minh City, Vietnam, November 18-20, 2019, Proceedings, 2019. p. 177–188. https://doi.org/10.1007/978-3-030-34980-6_20.
- Kowsari K, Meimandi KJ, Heidarysafa M, Mendu S, Barnes LE, Brown DE. Text Classification Algorithms: A Survey. Information. 2019;10(4):150. https://doi.org/10.3390/info10040150.
- 36 Akhtar MS, Ekbal A, Cambria E. How Intense Are You? Predicting Intensities of Emotions and Sentiments using Stacked Ensemble [Application Notes]. IEEE Comput Intell Mag. 2020;15(1):64–75. https://doi.org/10.1109/MCI.2019.2954667.
- 37 Graff M, Miranda-Jiménez S, Tellez ES, Moctezuma D. Evomsa: A multilingual evolutionary approach for sentiment analysis [application notes]. IEEE Comput Intell Mag. 2020;15(1):76– 88. https://doi.org/10.1109/MCI.2019.2954668.
- Yim J, Joo D, Bae J, and Kim J. A Gift from Knowledge Distillation: Fast Optimization, Network Minimization and Transfer Learning. In 2017 IEEE Conference on Computer Vision and Pattern Recognition, CVPR 2017, Honolulu, HI, USA, July 21-26, 2017. p. 7130–7138. https://doi.org/10.1109/CVPR.2017.754.
- Xing FZ, Pallucchini F, Cambria E. Cognitive-Inspired Domain Adaptation of Sentiment Lexicons. Information Processing & Management. 2019;56(3):554–564. Available from: https://doi. org/10.1016/j.ipm.2018.11.002.

- 40 Murtadha A, Qun C, Li Z. Constructing Domain-Dependent Sentiment Dictionary for Sentiment Analysis. Neural Comput Applic. 2020;32:14. https://doi.org/10.1007/s00521-020-04824-8.
- 41 Zhao C, Wang S, Li D. Multi-Source Domain Adaptation with Joint Learning for Cross-Domain Sentiment Classification. Knowl Based Syst. 2020;191:105254. https://doi.org/10.1016/j.knosys. 2019.105254.
- 42. Du C, Sun H, Wang J, Qi Q, and Liao J. Adversarial and Domain-Aware BERT for Cross-Domain Sentiment Analysis. In Proceedings of the 58th Annual Meeting of the Association for Computational Linguistics, ACL 2020, Online, July 5-10, 2020. p. 4019–4028. https://www.aclweb.org/anthology/2020.acl-main.370/.
- 43 Nguyen HL, Jung JJ. Social Event Decomposition for Constructing Knowledge Graph. Futur Gener Comput Syst. 2019;100:10–8. https://doi.org/10.1016/j.future.2019.05.016.
- 44 Camacho D, Luzón MV, Cambria E. New Trends and Applications in Social Media Analytics. Futur Gener Comput Syst. 2021;114:318–21. https://doi.org/10.1016/j.future.2020.08.007.
- 45 Cambria E, Hussain A. Sentic Computing. Cogn Comput. 2015;7(2):183–5. https://doi.org/10.1007/s12559-015-9325-0.
- 46. Jou B, Chen T, Pappas N, Redi M, Topkara M, and Chang S. Visual Affect Around the World: A Large-scale Multilingual Visual Sentiment Ontology. In Proceedings of the 23rd Annual ACM Conference on Multimedia Conference, MM '15, Brisbane, Australia, October 26 - 30, 2015. p. 159–168. https://doi. org/10.1145/2733373.2806246.
- Zhuang L, Schouten K, Frasincar F. SOBA: Semi-Automated Ontology Builder for Aspect-Based Sentiment Analysis. J Web Semant. 2020;60:100544. https://doi.org/10.1016/j.websem. 2019.100544.
- Lehmann J, Isele R, Jakob M, Jentzsch A, Kontokostas D, Mendes PN, Hellmann S, Morsey M, van Kleef P, Auer S, Bizer C. DBpedia - A Large-Scale, Multilingual Knowledge Base Extracted from Wikipedia. Semantic Web. 2015;6(2):167–95. https://doi. org/10.3233/SW-140134.
- 49 Vrandecic D, Krötzsch M. Wikidata: A Free Collaborative Knowledgebase. Commun ACM. 2014;57(10):78–85. https://doi. org/10.1145/2629489.
- 50 Cavallari S, Cambria E, Cai H, Chang KC, Zheng VW. Embedding Both Finite and Infinite Communities on Graphs [Application Notes]. IEEE Comput Intell Mag. 2019;14(3):39–50. https://doi. org/10.1109/MCI.2019.2919396.
- Ghosal D, Hazarika D, Roy A, Majumder N, Mihalcea R, and Poria S. KinGDOM: Knowledge-Guided DOMain Adaptation for Sentiment Analysis. In Proceedings of the 58th Annual Meeting of the Association for Computational Linguistics, ACL 2020, Online, July 5-10, 2020. p. 3198–3210. https://www.aclweb.org/ anthology/2020.acl-main.292/.
- Bijari K, Zare H, Kebriaei E, Veisi H. Leveraging Deep Graphbased Text Representation for Sentiment Polarity Applications. Expert Systems with Applications. 2020;144:113090. https://doi. org/10.1016/j.eswa.2019.113090.
- 53. Liu W, Zhou P, Zhao Z, Wang Z, Ju Q, Deng H, and Wang P. K-BERT: Enabling Language Representation with Knowledge Graph. In The Thirty-Fourth AAAI Conference on Artificial Intelligence, AAAI 2020, The Thirty-Second Innovative Applications of Artificial Intelligence Conference, IAAI 2020, The Tenth AAAI Symposium on Educational Advances in Artificial Intelligence, EAAI 2020, New York, NY, USA, February 7-12, 2020. p. 2901–2908. https://aaai.org/ ojs/index.php/AAAI/article/view/5681.
- 54 Seifollahi S, Shajari M. Word Sense Disambiguation Application in Sentiment Analysis of News Headlines: an Applied Approach to FOREX Market Prediction. J Intell Inf Syst. 2019;52(1):57–83. https://doi.org/10.1007/s10844-018-0504-9.

- Ji S, Pan S, Cambria E, Marttinen P, and Yu PS. A Survey on Knowledge Graphs: Representation, Acquisition and Applications. CoRR, abs/2002.00388, 2020. https://arxiv.org/abs/2002.00388.
- 56 Weichselbraun A, Gindl S, Fischer F, Vakulenko S, Scharl A. Aspect-Based Extraction and Analysis of Affective Knowledge from Social Media Streams. IEEE Intell Syst. 2017;32(3):80–8. https://doi.org/10.1109/MIS.2017.57.
- 57. Weichselbraun A, Kuntschik P, and Brasoveanu AMP. Name Variants for Improving Entity Discovery and Linking. In 2nd Conference on Language, Data and Knowledge, LDK 2019, May 20-23, 2019, Leipzig, Germany 2019. p. 14:1–14:15. https://doi. org/10.4230/OASIcs.LDK.2019.14.
- Scharl A, Hubmann-Haidvogel A, Göbel MC, Schäfer T, Fischl D, and Nixon LJB. Multimodal Analytics Dashboard for Story Detection and Visualization. In Video Verification in the Fake News Era, Springer 2019. p. 281–299. https://doi.org/10.1007/978-3-030-26752-0_10.
- Weichselbraun A, Scharl A, and Lang H. Knowledge Capture from Multiple Online Sources with the Extensible Web Retrieval Toolkit (eWRT). In Proceedings of the 7th International Conference on Knowledge Capture, K-CAP 2013, Banff, Canada, June 23-26, 2013. p. 129–132. https://doi.org/10.1145/2479832.2479861.
- Weichselbraun A, and Süsstrunk N. Optimizing Dependency Parsing Throughput. In KDIR 2015 - Proceedings of the International Conference on Knowledge Discovery and Information Retrieval, part of the 7th International Joint Conference on Knowledge Discovery, Knowledge Engineering and Knowledge Management (IC3K 2015), Volume 1, Lisbon, Portugal, November 12-14, 2015. p. 511–516. https://doi.org/10.5220/0005638905110516.
- 61. Choi JD, Tetreault JR, and Stent A. It Depends: Dependency Parser Comparison Using A Web-based Evaluation Tool. In Proceedings of the 53rd Annual Meeting of the Association for Computational Linguistics and the 7th International Joint Conference on Natural Language Processing of the Asian Federation of Natural Language Processing, ACL 2015, July 26-31, 2015, Beijing, China, Volume 1: Long Papers 2015. p. 387–396. https:// doi.org/10.3115/v1/p15-1038.
- 62. Chatterjee A, Narahari KN, Joshi M, and Agrawal P. SemEval-2019 Task 3: EmoContext Contextual Emotion Detection in Text. In Proceedings of the 13th International Workshop on Semantic Evaluation, SemEval@NAACL-HLT 2019, Minneapolis, MN, USA 2019, p. 39–48. https://doi. org/10.18653/v1/s19-2005.
- 63 Sarker A, Belousov M, Friedrichs J, Hakala K, Kiritchenko S, Mehryary F, Han S, Tran T, Rios A, Kavuluru R, de Bruijn B, Ginter F, Mahata D, Mohammad SM, Nenadic G, Gonzalez-Hernandez G. Data and systems for medication-related text classification and concept normalization from twitter: insights from the social media mining for health (SMM4H)-2017 shared task. J Am Med Infor Assoc. 2018;25(10):1274–83. https://doi. org/10.1093/jamia/ocy114.

- Klinger R, Clercq OD, Mohammad SM, and Balahur A. IEST: WASSA-2018 Implicit Emotions Shared Task. CoRR, abs/1809.01083, 2018. http://arxiv.org/abs/1809.01083.
- 65. Saravia E, Liu HT, Huang Y, Wu J, and Chen Y. CARER: Contextualized Affect Representations for Emotion Recognition. In Proceedings of the 2018 Conference on Empirical Methods in Natural Language Processing, Brussels, Belgium, October 31 - November 4, 2018. p. 3687–3697. https://doi.org/10.18653/v1/d18-1404.
- 66. Poria S, Hazarika D, Majumder N, Naik G, Cambria E, and Mihalcea R. MELD: A multimodal multi-party dataset for emotion recognition in conversations. In Proceedings of the 57th Conference of the Association for Computational Linguistics, ACL 2019, Florence, Italy, July 28- August 2, 2019, Volume 1: Long Papers 2019. p. 527–536. https://doi.org/10.18653/v1/p19-1050.
- Ettinger A. What BERT is Not: Lessons from a New Suite of Psycholinguistic Diagnostics for Language Models. Transactions of the Association for Computational Linguistics. 2020;8:34–48. https://transacl.org/ojs/index.php/tacl/article/view/1852.
- Khandelwal A, and Sawant S. NegBERT: A Transfer Learning Approach for Negation Detection and Scope Resolution. In Proceedings of The 12th Language Resources and Evaluation Conference, LREC 2020, Marseille, France, May 11-16, 2020. p. 5739–5748. https://www.aclweb.org/anthology/2020.lrec-1.704/.
- Lengauer G, Esser F, Berganza R. Negativity in Political News: A Review of Concepts. Operationalizations and Key Findings. Journalism. 2012;13(2):179–202. https://doi.org/10.1177/ 2F1464884911427800.
- Hewitt J and Manning CD. A Structural Probe for Finding Syntax in Word Representations. In Proceedings of the 2019 Conference of the North American Chapter of the Association for Computational Linguistics: Human Language Technologies, NAACL-HLT 2019, Minneapolis, MN, USA, June 2-7, 2019, Volume 1 (Long and Short Papers) 2019. p. 4129–4138. https://doi.org/10.18653/ v1/n19-1419.
- Maudslay RH, Valvoda J, Pimentel T, Williams A, and Cotterell R. A tale of a probe and a parser. In Proceedings of the 58th Annual Meeting of the Association for Computational Linguistics, ACL 2020, Online, July 5-10, 2020. p. 7389–7395. https://www. aclweb.org/anthology/2020.acl-main.659/.
- Vig J. A Multiscale Visualization of Attention in the Transformer Model. In *P*roceedings of the 57th Conference of the Association for Computational Linguistics, ACL 2019, Florence, Italy, July 28 - August 2, 2019, Volume 3: System Demonstrations, 2019. p. 37–42. https://doi.org/10.18653/v1/p19-3007.